Rotary Gas Expanders Radiusl

Case Study #3

Bakken Basin — Oil Well
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Names Units Liquids to Pipeline
VLVE-101 MIX-101
C2(Std Liquid Volumetric Flow ) |gal/d 916.01 JT_C2_recovery |6.9801 |% 15
C3(Std Liquid Volumetric Flow ) |gal/d 1980 JT_C3_recovery |23.805 |% ¢ Liquids to Pipeline -
iC4(Std Liquid Volumetric Flow) |gal/d 458.91 JT_iCA_recovery |36.387 % Flash sep >
nC4(Std Liquid Volumetric Flow ) |gal/d 1571.7 JT_nC4_recovery |38.142 |% . !
iC5(Std Liquid Volumetric Flow) |galid 401.35 JT_iC5_recovery [35.723 [% piitigaidiVolimeticlhlowjior2o8oalld
nC5(Std Liquid Volumetric Flow ) |gal/d 632.81 JT_nC5_recovery |32.899 |% True Vapor Pressure 223 psig
nC6(Std Liquid Volumetric Flow ) |gal/d 671.08 JT_C6_recovery |17.495 |%




With Expander
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Expander power = -58.65 HP
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Units Liquids to Pipeline
gal/d 1415.3 C2 recovery [10.8 (% Std Liquid VVolumetric Flow 8930.5 gal/d
galid 3055.2 C3_recovery |36.7 (% True Vapor Pressure 228 psig
gal/d 609.14 iC4_recovery |48.3 |%
gald 1944 nC4_recovery [47.2 |%
galid 4433 iC5_recovery |39.5 |%
gal/d 683.04 nC5_recovery [35.5 |%
gal/d 682.49 C6__recovery |17.8 (%

nC6(Std Liquid Volumetric Flow )




Assuming 2.8 mmscfd flow

Radiusl

Williston Basin — Oil Well

JT Skid with Expander Difference Benefit

Measured
'(I'Itz-:-)mperature 14.9 -12 27 deg Condensate
NGL's

6,620 9,705 47% SSS

(gpd) i
Electricity 0 43.7 o 257 metric tons
(kW) (CO,) prevented*
Heating Valve 0
(BTU/ftA3) 1,156 1,109 4%

* Greenhouse Gas Equivalencies Calculator | US EPA



https://www.epa.gov/energy/greenhouse-gas-equivalencies-calculator#results
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